### **Cultural Diversity and Ethnic Minority Psychology**

## The Pandemic and Social Experience: For Whom Did Discrimination and Social Isolation Increase?

Thomas E. Fuller-Rowell, Olivia I. Nichols, Mona El-Sheikh, Anthony L. Burrow, Anthony D. Ong, and Carol D. Ryff Online First Publication, September 12, 2022. http://dx.doi.org/10.1037/cdp0000561

#### CITATION

Fuller-Rowell, T. E., Nichols, O. I., El-Sheikh, M., Burrow, A. L., Ong, A. D., & Ryff, C. D. (2022, September 12). The Pandemic and Social Experience: For Whom Did Discrimination and Social Isolation Increase?. *Cultural Diversity and Ethnic Minority Psychology*. Advance online publication. http://dx.doi.org/10.1037/cdp0000561



Society for the Psychological Study of Culture, Ethnicity, and Race

https://doi.org/10.1037/cdp0000561

# The Pandemic and Social Experience: For Whom Did Discrimination and Social Isolation Increase?

Thomas E. Fuller-Rowell<sup>1</sup>, Olivia I. Nichols<sup>1</sup>, Mona El-Sheikh<sup>1</sup>,

Anthony L. Burrow<sup>2</sup>, Anthony D. Ong<sup>2</sup>, and Carol D. Ryff<sup>3</sup>

<sup>1</sup> Department of Human Development and Family Science, Auburn University

<sup>2</sup> Department of Psychology, Cornell University <sup>3</sup> Department of Psychology, University of Wisconsin–Madison

Department of 1 sychology, Oniversity of Wisconsin-Waarson

**Objective:** The objective of this study was to examine whether experiences of discrimination have increased during the pandemic, particularly among negatively stigmatized racial/ethnic groups, and whether such experiences have exacerbated feelings of social isolation. **Method:** Discrimination and social isolation were assessed before and during the pandemic in a sample of 263 Black and White young adults attending a large, predominantly White 4-year research university in the Southeastern region of the United States (52% Black, 48% White, 53% female, mean age = 19.2). **Results:** Increases in discrimination were evident among Black but not White participants. Black participants also reported greater increases in social isolation than White participants, and changes in discrimination partially mediated the emergent racial disparity in social isolation. **Conclusions:** Findings are consistent with theoretical perspectives on discrimination during times of stress and suggest the need for broader attention to the impact of the pandemic on unfair treatment of stigmatized groups.

#### Public Significance Statement

The results of this study suggest notable shifts in experiences of discrimination and social isolation during the COVID-19 pandemic. In particular, the findings indicate that Black young adults experienced increases in discrimination, which exacerbated Black–White racial disparities in social isolation. Determining the degree to which the reported changes are enduring, evident in other demographic groups, and of longer term developmental and public health significance will be important next steps to inform a continuing pandemic response, and responses to future societal-level stressful events.

Keywords: COVID-19 pandemic, racial disparities, social isolation, experiences of discrimination, young adults

Supplemental materials: https://doi.org/10.1037/cdp0000561.supp

Burdens of the COVID-19 pandemic have been enormous (Beaney et al., 2020) and have been disproportionately borne by those who were already disadvantaged (Serkez, 2021). Rates of infection and mortality have been higher among racial/ethnic minority and socioeconomically disadvantaged groups (Woolf et al., 2021), as have rates of job, housing, and financial loss (Perry et al., 2021). A burgeoning literature now also addresses the deleterious impact of the early phases of the pandemic on experiences of social isolation and psychological distress (Ernst et al., 2022; Ettman et al., 2020; Peng & Roth, 2022; Pierce et al., 2020; Salari et al., 2020). Few studies, however, have considered the role that unfair treatment and discrimination may play in shaping adverse psychosocial consequences of the pandemic.

Have experiences of discrimination increased? Are such increases happening to a greater degree among marginalized racial/ethnic groups? And if such changes are occurring, what are their consequences for social isolation? These questions are of theoretical and

Thomas E. Fuller-Rowell D https://orcid.org/0000-0002-1447-1175

This work was supported by the National Heart, Lung, and Blood Institute (Grant 1R15HL140504-01A1); and the United States Department of Agriculture, National Institute of Food and Agriculture (Hatch project 1003947). The funding sources had no role in the study design or in the collection, analysis, interpretation, or write up of the data. The authors declare that they have no conflict of interest.

Thomas E. Fuller-Rowell played lead role in conceptualization, data curation, formal analysis, funding acquisition, investigation, methodology, supervision, writing of original draft, and writing of review and editing. Olivia I. Nichols played supporting role in conceptualization, data curation, formal analysis, investigation, writing of original draft, and writing of review and editing. Mona El-Sheikh played supporting role in conceptualization, funding acquisition, methodology, and writing of review and editing. Anthony L. Burrow played supporting role in conceptualization and writing of review and editing. Anthony D. Ong played supporting role in conceptualization and writing of review and editing. Carol D. Ryff played supporting role in conceptualization, writing of original draft, and writing of review and editing.

Correspondence concerning this article should be addressed to Thomas E. Fuller-Rowell, Department of Human Development and Family Science, Auburn University, 203 Spidle Hall, Auburn, AL 36849, United States. Email: tef0005@auburn.edu

practical importance in illuminating the impact of the COVID-19 pandemic and advancing understanding of unfolding social inequities, but as yet have been largely unaddressed by empirical research (Rosenfeld et al., 2022; Stok et al., 2021). Two notable exceptions include work documenting adverse treatment of Asian Americans during the pandemic (Chen et al., 2020; Wu et al., 2021), and research showing higher levels of coronavirus-specific discrimination among racial/ethnic minority groups as compared to Whites (Liu et al., 2020). To our knowledge, few, if any, studies have examined whether general perceived discrimination (i.e., unfair treatment due to any cause) has increased during the pandemic from prepandemic levels or its potential consequences for social isolation. The significance of these topics is supported by several established literature.

Foundational theories in social cognitive psychology suggest that stress, fatigue, and related cognitive depletion are associated with increased reliance on stereotypes (Bodenhausen, 1990; Tajfel & Turner, 2001). For example, studies show that as stress and negative emotions increase, so do responses that reflect oversimplified cognitive schema or fixed ideas about social categories such as race (Bodenhausen, 1993; Sherman & Frost, 2000). A related line of research also indicates that under conditions of pathogen threat—the perception of heightened infectious disease risk—authoritarian attitudes and right-wing conservative perspectives increase (Moran et al., 2021; Pazhoohi & Kingstone, 2021). Some evidence also suggests that experimental priming of pathogen threat heightens prejudice toward unfamiliar out groups (Faulkner et al., 2004; Lu et al., 2021), and increases anti-Black bias (O'Shea et al., 2020).

Another related line of research indicates that discrimination may also increase under conditions of resource scarcity, such as during recessions (Anderson et al., 2020; Bianchi et al., 2018; Krosch et al., 2017). Collectively, these literatures suggest that because the pandemic and related changes to daily life may have (a) been experienced as stressful, (b) induced competition for resources, and (c) increased perceptions of disease risk—prejudice and reliance on heuristic processing strategies (i.e., habitual responses based on stereotypes) are likely to have increased. Although experiences of discrimination and unfair treatment may increase among all demographic subgroups under these conditions, increases in experiences of discrimination are likely to be most pronounced among racial/ethnic minority groups who have been historically marginalized and are more likely to be the targets of negative stigmatization (Blessum et al., 1998; Kusche & Barker, 2019).

As noted above, there is substantial evidence that the COVID-19 virus has had a greater impact in minority communities. For example, rates of infection and mortality have been more pronounced in Black communities than among Whites in the United States (Woolf et al., 2021). Reasons for these disparities have been detailed elsewhere (Garcia et al., 2021; Marmot & Allen, 2020; Njoku, 2021) and relate to structural racism and resulting race differences in resources and exposures (e.g., within neighborhood and work contexts) that undermine health within minority communities and increase Black Americans' vulnerability to infection and preexisting conditions that elevate risk of COVID-19 mortality (National Academies of Sciences, Engineering, & Medicine, 2021). Substantial media attention given to race differences in infection early in the pandemic may have heightened negative feelings toward minority communities and led to increased unfair treatment toward those depicted as having high rates of viral

infection and mortality (Chen et al., 2020; Su et al., 2020). For example, disturbing increases in violence and hate crimes directed toward Asian Americans were documented (Gover et al., 2020; Pan et al., 2021; Wu et al., 2021). However, the degree to which experiences of discrimination may have increased within other racial/ethnic minority communities has received less attention, despite these communities frequently being depicted in the media as having higher rates of COVID-19 infection.

Social isolation, commonly defined as a lack of available social connections (Leigh-Hunt et al., 2017; Steptoe et al., 2013), is clearly of concern during the pandemic. Social distancing guidelines in the United States were implemented in March 2020, with substantial consequences for everyday patterns of social interaction (Hsiang et al., 2020). As a result, many have experienced increases in social isolation across this period (Ernst et al., 2022; Hamza et al., 2021). Additionally, less advantaged segments of the population are likely to have been more vulnerable to these increases. For example, due to a disproportionate burden of stressful experiences during the pandemic (e.g., family financial strain, job loss, death of a family member) and poorer access to stable internet connections and computer resources, Black Americans may have been less well situated to access and maintain their existing social networks (Fairlie, 2007; Perry et al., 2021). Furthermore, if experiences of discrimination increased more among Black Americans, this may also have led to greater increases in social isolation. By considering changes in social isolation alongside discrimination, the present study directly addresses the role that discrimination might play in shaping changes in social isolation during the pandemic and differences in these changes between Black and White Americans.

Of particular methodological importance is the need for comparative assessments of discrimination and loneliness before and during the pandemic. One such opportunity pertained to a cohort of African American (Black) and European American (White) young adult college students, who were assessed prior to the pandemic (during the 2018–2019 academic year) and again during the early months of the pandemic in the United States. Young adults are an important group in which to address the impact of the pandemic for several reasons: (a) early adulthood is a pivotal time in the life course for negotiating many life challenges and uncertainties, including establishing career and relational identities that may set the course of subsequent development and health (Arnett et al., 2014; Schwartz, 2016); (b) there is evidence that social disruptions and related detrimental effects of the pandemic may have been greater among young adults (O'Connor et al., 2021; Pierce et al., 2020); (c) young adulthood is a particularly salient developmental period for the onset of psychopathology, with studies estimating that nearly three quarters of all mental disorders start before the mid-20s (Kessler et al., 2007; Solmi et al., 2022); and (d) early adulthood is particularly fraught for Black young adults in the United States, who experience more interpersonal racism and discrimination than other groups and must grapple with the significance and meaning of an identity that is often devalued in society (Fuller-Rowell et al., 2011; Hope et al., 2015). Thus, understanding how this age group was impacted by the pandemic is of particular importance to policy discussions and debates surrounding how the adverse economic and health consequences of the pandemic-and inequities in the distribution of these consequences-can be mitigated.

College contexts, though relatively advantaged socioeconomically, are an important stage on which to examine unfolding inequities of the pandemic among young adults. Black college students are known to experience disproportionate stress exposures, which come with documented mental and physical health burdens (Fuller-Rowell et al., 2017; Priest et al., 2013), and undermine some of the health benefits typically obtained from higher educational attainment (Assari, 2018; Fuller-Rowell et al., 2015). These troubling realities suggest that close monitoring of unfolding inequalities within contexts of higher education are of paramount importance.

Based on the above lines of evidence, we postulate (a) that, for both Black and White students, experiences of discrimination and social isolation will have increased during the early phase of the pandemic from prepandemic levels; (b) that increases in discrimination will have been greater among Black than White students; and (c) that greater increases among Black students will partially mediate the widening of racial disparities in social isolation.

#### Method

#### Participants and Design

Participants were 263 first- and second-year undergraduate college students (52% Black; 48% White; 53% female;  $M_{age}$  = 19.21 years, SD = 1.01), attending a large selective 4-year research university in the Southeastern region of the United States with a predominantly White student body of more than 20,000. Informed consent was obtained from all participants, and all human subjects' activities were conducted in compliance with an appropriate internal review board. Approximately equal numbers of African American (Black) and European American (White) students were contacted during the academic year, between September 2018 and April 2019 (T1), using official university records to identify demographic characteristics of students. This approach allowed for adequate sample sizes to examine differences between Black (N = 137, 58% female) and White (N = 127, 48% female) students. Inclusion criteria were (a) being an undergraduate student aged 18-25 and (b) having a race of Black or White in student records (Fuller-Rowell et al., 2021). White students who were first generation (did not have a parent who had graduated from a 4-year college) were oversampled to reduce potential confounding by socioeconomic status. In total, 468 Black and 608 White students (33% first generation in both groups) were sent information (pamphlet and letter) about the study via mail (to their local and permanent addresses) and emails (to their university email address). The study was described as seeking to understand factors in students' lives that influence their health. The final sample contained 33% first-generation students in both racial/ethnic groups. Median household income was \$100,000 (SD = 68,000), 20% grew up in families that did not own their own home for >50% of their childhood, 8% had household incomes less than 150% of the federal poverty line, and 34% had family incomes less than 300% of the poverty line. For reference, 300% of the federal poverty line for a family of four in 2019 was \$77,250, and 43% of the Black sample and 24% of the White sample fell below that threshold. Participants were compensated \$70 for the initial assessment, during which a variety of psychosocial and health measures were administered. At the baseline assessment, 10% of students lived at home with family and remaining 90% living in on- or off-campus housing.

A follow-up assessment was conducted 1.27 years later (SD = 0.18), between April 27 and June 12, 2020 (T2), during the first

wave of the COVID-19 pandemic in the United States. At the follow-up, 61% of students had experienced a change in housing, and 98% of those indicated that the change was to have moved back home with family. Participants were emailed a Qualtrics survey and compensated \$25 for their participation. Of the 263 participants in the T1 sample, 76% (N = 200) participated in the T2 survey (56%) female, 49% Black). Within each time point, there was less than 1%of missing data at T1 and less than 2% at T2. Missing data were addressed using full-information maximum likelihood estimation (FIML). Levels of missingness over time, and within each time point, were acceptable for FIML estimation (Enders, 2010). Using FIML estimation, including variables associated with attrition in the models, helps to correct for bias that may be introduced by selective attrition and allows for the full T1 sample (N = 263) to be included in all analyses (Acock, 2005; Enders, 2013). There was a statistical trend such that those lost to attrition between T1 and T2 were slightly more likely to occur among Black (p = .074) and female (p = .058) respondents, respectively. Sensitivity analyses focusing only on the participants with complete data at both time points (i.e., using listwise deletion) yielded no differences in the pattern of findings reported.

#### Measures

#### **Everyday Discrimination**

Perceptions of discrimination were assessed at both time points using the Everyday Discrimination (EVD) Scale (Williams et al., 1997). The scale is a widely used measure to assess perceived discrimination with numerous studies documenting internal consistency (Krieger et al., 2005; Taylor et al., 2004) and associations with health (Williams & Mohammed, 2009). For each of the nine items, participants indicated how often experiences of unfair treatment occurred in their day-day-day life (e.g., "you are treated with less courtesy than other people," "people act as if they are afraid of you," and "people act as if they think you are dishonest"). Items were rated on a 7-point scale (1 = never to 7 = several times a day) and mean scored at both time points ( $\alpha_{T1} = .878$ ;  $\alpha_{T2} = .904$ ). After completing the measure, a follow-up question also asked participants to indicate the main reason for the experiences reported. Descriptive statistics for reasons (attributions) reported at each time point are provided in the online Supplemental Materials.

#### Social Isolation/Support

At both time points, social isolation/support was assessed using the Interpersonal Support Evaluation List–12 (ISEL-12; Cohen et al., 1985). The measure is widely used in health research, prior studies have documented internal consistency, and items have been shown to load well on a single factor (Merz et al., 2014). Convergent validity with other measures of social support/isolation, and predictive validity with mental and physical health, have also been established (Dinenberg et al., 2014; Payne et al., 2012). The scale is organized into three dimensions with four items assessing each dimension: (a) *appraisal* (having people to turn to when you have a personal problem), for example, "I feel that there is no one I can share my most private worries and fears with" and "If a family crisis arose, it would be difficult to find someone who could give me good advice about how to handle it"; (b) *belonging* (having people to do recreational things with), for example, "If I wanted to have lunch with someone, I could easily find someone to join me" and "If I wanted to go on a trip for a day (e.g., to the country or mountains), I would have a hard time findings someone to go with me"; and (c) tangible support (having people available who are willing to help with practical tasks), for example, "If I were sick, I could easily find someone to help me with my daily chores" and "If I was stranded 10 miles from home, there is someone I could call who could come and get me." For each of the items, respondents indicate how true or false the statement is on a 4-point scale from definitely false to definitely true (coded 0-3). A sum score of all items is taken as an overall index of social isolation/support. For the present study, items were coded such that higher scores indicate greater social isolation or less support ( $\alpha_{T1} = .837; \alpha_{T2} = .790$ ), and thus, for simplicity, we refer to the measure as social isolation throughout. Because the majority of items included in the ISEL-12 directly imply social isolation, we contend that it is reasonable to refer to the measure as such. Although some prior research has distinguished social support from social isolation (Gable & Bedrov, 2022), support and isolation are closely linked constructs empirically and conceptually, for this study, we do not distinguish them.

#### Demographic Covariates

Both race and gender were coded from student records obtained from the university (White = 0, Black = 1; female = 0, male = 1) and confirmed via self-reports. Age was assessed at time T1 and coded in years. Parent education and household income were measured as indicators of family socioeconomic status. Participants were asked to report each parent's level of education on an 11-point scale ranging from no school or some grade school (coded as 1) to PhD, EDD, MD, DDS, LLB, LLD, JD, or other professional degree (coded as 11). Parent education levels were averaged to create a single-parent education score. Household income was calculated by summing participant responses of individual parent income in the primary household, measured on a scale with 30 possible categories  $(\$1 \le \$5,000 - \$30 \ge \$500,000)$ . Family household income was then divided by the corresponding U.S. Census 2010 poverty line based on family size to calculate a family-adjusted measure of household income using an income-to-needs ratio (Fuller-Rowell et al., 2012).

#### **Analysis Plan**

Our analytic approach followed three strps. First, unadjusted mean differences in discrimination and social isolation were plotted for each racial group and compared using t tests. Second, a series of longitudinal structural equation models (SEM) were estimated in Mplus Version 8.4 to examine predictors of change in each measure of interest (Muthen & Muthen, 1998; Newsom et al., 2013). Models were estimated separately for social isolation and everyday discrimination before fitting a combined model. To allow for estimates to be interpreted with respect to change over time in T1 SD units, T2 measures were standardized by subtracting the T1 mean and dividing by the T1 SD (Cohen et al., 2003). Model 1 included only one predictor variable, the T1 autoregressive control, which was zscored. In these models, the intercept can be interpreted as the mean change in the outcome variable (in T1 SD units) for the full sample between the two assessments. To examine race differences in change adjusting for other demographic variables, Model 2 then

added race, age, gender, parent education, and household income as predictors. Age, parent education, and household income were z scored for clarity of interpretation.

To account for measurement error in assessment of social isolation and everyday discrimination, single indicator latent variables were used for each scale score and residual variances were fixed using the formula  $(1 - \text{Reliability}) \times \text{Sample Variance (Choi et al.,})$ 2011; Hayduk, 1987). Cronbach's  $\alpha$  was used to estimate reliability at each time point. Using this approach, model parameter estimates are equivalent to lagged regression models, with the only key difference being that measurement error was specified rather than assumed to be zero (Cole & Preacher, 2014; Newsom et al., 2013). Last, a final structural equation model was estimated using the same modeling strategy to examine indirect effects of race on social isolation through everyday discrimination. Indirect effects were estimated using the product of coefficients method with biascorrected bootstrapped confidence intervals (Haves & Scharkow, 2013). Two waves of data offer an initial step for mediation analyses and provide important strengths over cross-sectional analyses (Cole & Maxwell, 2003; Roth & MacKinnon, 2011). Additional analyses probed methodological assumptions, considered possible alternative explanations for the findings, and examined possible moderation by race, sex, and socioeconomic status.

#### Results

Correlations and descriptive statistics for the full sample are shown in Table 1. Participants reported significantly more social isolation and discrimination at T2 than at T1. In the full sample, social isolation increased between T1 and T2 by 2.08 units (SE =0.63, p = .001; 0.31 SD units) and everyday discrimination increased by 0.20 units (SE = 0.09, p = .024; 0.23 SD units). Experiences of discrimination were positively correlated with social isolation at both T1 (r = .27, p < .001) and T2 (r = .34, p < .001). Descriptive statistics, shown separately for Black and White participants, are reported in Table 2. Black participants had lower family income than White participants. Black participants reported more experiences of discrimination at T2 but not T1, and higher levels of social isolation at both time points than White participants. These findings are further elucidated below.

#### **Changes in Everyday Discrimination**

Figure 1 shows unadjusted means of everyday discrimination by race at *T*1 and *T*2. The unadjusted race differences in everyday discrimination were not significant at the *T*1 assessment (p = .187). However, Black students had substantially higher levels of discrimination than White students at the *T*2 assessment (*M* difference = 0.61 *SD* units, *SE* = 0.16, p < .001).

Regression models were estimated in Mplus to examine the magnitude of changes in everyday discrimination and to consider race differences in change across the two time points, adjusting for age, gender, and socioeconomic status (parent education and household income). Results from Model 1 indicated that discrimination at *T*1 was a significant predictor of discrimination at *T*2, explaining 38% of the *T*2 variance (B = 0.73, SE = 0.08, p < .001). The Model 1 intercept also indicated that, in the full sample, the average increase in everyday discrimination between *T*1 and *T*2 was 0.23 *SD* units (SE = 0.08, p = .004). In Model 2, race was added as a predictor and

 Table 1

 Correlations and Descriptive Statistics for T1 and T2 Variables

correlations and Descriptive Statistics for 11 and 12 variables								
Study variable	1	2	3	4	5	6	7	8
1. Race (Black)	_							
2. Sex (female)	0.10	_						
3. Parent education	0.02	-0.02	_					
4. Income-to-needs ratio	-0.26	-0.07	0.30	_				
5. T1 discrimination	0.08	0.07	0.07	0.03	_			
6. T2 discrimination	0.27	0.05	0.01	-0.04	0.56	_		
7. T1 social isolation	0.13	0.01	-0.03	-0.07	0.27	0.33		
8. T2 social isolation	0.21	-0.17	-0.01	0.01	0.16	0.34	0.45	_
<i>M</i> (or %)	(52.1)	(53.6)	7.43	4.54	1.94	2.14	7.46	9.54
SD			2.04	3.03	0.85	0.98	6.68	6.68

*Note.*  $T = \text{time. Correlations and descriptive statistics were generated using full-information maximum likelihood in Mplus allowing the full sample to be included (<math>N = 263$ ). Statistically significance correlations (p < .05) are shown in bold font.

was found to explain an additional 6% of the variance in everyday discrimination at *T*2 after adjusting for covariates. The magnitude of this effect indicated that Black students experienced a 0.55 *SD* unit (*SE* = 0.14, p < .001) greater increase in everyday discrimination scores between *T*1 and *T*2 than White students. Specifically, Black students had a 0.52 *SD* unit (*SE* = 0.11, p < .001) increase in everyday discrimination from *T*1 to *T*2, whereas White students had no significant change in everyday discrimination from *T*1 to *T*2 (*B* = -0.03, *SE* = 0.10, p = .773).

#### **Changes in Social Isolation**

Figure 2 shows mean levels of social isolation by race at *T*1 and *T*2. Black students reported significantly higher levels of social isolation than White students at both *T*1 (*M* difference = 0.27 *SD* units, SE = 0.12, p = .027) and *T*2 (*M* difference = 0.44 *SD* units, SE = 0.14, p = .002). The magnitude of the race difference in social isolation at *T*2 was nearly double that of *T*1.

Regression models were again estimated in Mplus to examine changes in social isolation and to consider race differences in change across the two time points, adjusting for age, gender, parent education, and household income. Results from Model 1 indicated that social isolation at *T*1 was a significant predictor of isolation at *T*2, explaining 32% of the variance after adjusting for demographic covariates (B = 0.55, SE = 0.08, p < .001). The Model 1 intercept indicated that, in the full sample, the average increase in social isolation between *T*1 and *T*2 was 0.31 *SD* units (SE = 0.07, p < .001).

#### Table 2

Descriptive	Statistics	for	Black	and	White	Participants
-------------	------------	-----	-------	-----	-------	--------------

	White students	Black students		
Variable	$M \pm SD$ (%)	$M \pm SD$ (%)	р	
Sex (female)	48.4%	58.4%	.103	
Parent education	$7.39 \pm 2.06$	$7.47 \pm 2.03$	.752	
Income-to-needs ratio	$5.37 \pm 3.66$	$3.81 \pm 2.12$	<.001	
T1 discrimination	$1.87 \pm 0.85$	$2.01 \pm 0.85$	.184	
T2 discrimination	$1.89 \pm 0.88$	$2.40 \pm 1.02$	<.001	
T1 social isolation	$6.50 \pm 5.79$	$8.33 \pm 7.33$	.024	
T2 social isolation	$8.10 \pm 6.72$	$11.06 \pm 6.36$	.002	

*Note.* T = time. Independent samples t tests were used to test differences in means between Black (n = 137) and White (n = 127) participants.

In Model 2, race was added as a predictor and found to explain an additional 8% of the variance in social isolation at *T*2 after adjusting for covariates. Black students experienced a 0.38 *SD* unit greater increase in social isolation than White students (SE = 0.13, p = .004). More specifically, Black students had a 0.53 *SD* unit (SE = 0.09, p < .001) increase in social isolation from *T*1 to *T*2, whereas White students had no significant change (B = 0.15, SE = 0.09, p = .096). Male students experienced a 0.41 *SD* unit greater increase in social isolation than females (SE = 0.13, p = .001).

#### Discrimination as a Mediator of Race Differences in Social Isolation Over Time

Structural equation modeling was used to examine whether changes in everyday discrimination mediated race differences in social isolation across the two time points (Figure 3). Adjusting for autoregressive and cross-lagged associations (Cole & Maxwell, 2003), everyday discrimination at T2 was associated with a 0.22 SD unit increase in reports of social isolation at T2 (SE = 0.08, p =.010). As hypothesized, a significant indirect effect of race on T2 social isolation through T2 discrimination was also evident (B =0.10, SE = 0.05, p = .032) and indicated that 27% of the total race difference in social isolation T2 (adjusting for social isolation at T1) was mediated by everyday discrimination (Figure 3). Autoregressive paths for both discrimination (B = 0.65, SE = 0.09, p < .001) and social isolation (B = 0.46, SE = 0.08, p < .001) were significant and moderate in magnitude. The cross-lagged association between social isolation at T1 and everyday discrimination at T2 (B = 0.21, SE = 0.09, p = .013) was also significant, whereas the association between discrimination at T1 and social isolation at T2 was not significant (B = -0.13, SE = 0.11, p = .223).

#### Supplemental Analyses

Whether socioeconomic background was associated with heightened experiences of discrimination was also examined. Results indicated that this was not the case, and that socioeconomic measures did not attenuate race effects or interact with race to influence experiences of discrimination. Otherwise put, analyses did not provide evidence that socioeconomic status moderated or mediated the reported race effects.

**Figure 1** *Mean Differences in Everyday Discrimination Between Black and White Students* 



*Note.* T = time. All scores are standardized (*z* scored); *T*2 scores were standardized by subtracting the *T*1 mean and dividing by the *T*1 standard deviation to allow for comparison of the two collection periods. Error bars indicate 95% confidence intervals. N = 263.

To probe normative developmental change as a potential alternative explanation for the findings, we also considered whether the magnitude of change in social isolation or discrimination was different for first-year college students than for more advanced students. Of the 263 students in the sample, 130 (49.4%) were in their first year at the baseline assessment (2018–2019), and the remaining 133 (50.6%) were more advanced undergraduate

#### Figure 2



Mean Differences in Social Isolation Between Black and White Students

*Note.* T = time. All scores are standardized (*z* scored); *T*2 scores were standardized by subtracting the *T*1 mean and dividing by the *T*1 standard deviation to allow for comparison of the two collection periods. Error bars indicate 95% confidence intervals. N = 263.

students. Results showed no differences between first-year students and more advanced students. Moderation analyses were also conducted to consider whether the strength of the association between discrimination and social isolation varied by race or sex. No significant differences were evident by race or sex.

Last, because the murder of George Floyd on May 25, 2020, and subsequent protests occurred near the end of our follow-up data collection period, sensitivity analyses were conducted to consider their possible impact on the results. A small number of participants (n = 17) completed the follow-up assessment on or after May 25. The pattern of findings and inference remained equivalent after removing these participants from the analyses.

#### Discussion

Despite established links between stress and discrimination, few studies have examined levels of discrimination experienced by Black Americans during the pandemic. Focusing on a fortuitous young adult college student sample with a two-wave longitudinal design, the present study examined changes in experiences of discrimination and social isolation from before to several months into the COVID-19 pandemic. Of particular interest were hypothesized differences in these changes between Black and White young adults.

Experiences of discrimination increased significantly for Black but not White young adults. For Black young adults, discrimination increased nearly half of a standard deviation across the two assessments, whereas White young adults showed no significant changes in reports of discrimination. Social isolation increased for both Black and White young adults, but the increase was significantly larger among Black young adults (.54 *SD* units vs. .14 *SD* units). The larger increase in social isolation in the Black sample was partially mediated by the increase in experiences of discrimination, with 27% of the race difference in social isolation being explained by discrimination (Figure 2). That is, increases in discrimination experienced by Black students partially mediated the emergent racial disparity in social isolation.

The findings have several implications for theory and practice. First, the findings add to and are congruent with separate established literature, showing that psychological distress has increased during the pandemic (Ettman et al., 2020; O'Connor et al., 2021), and that distress is linked to increased expressions of bias toward negatively stigmatized minority groups (Anderson et al., 2020; Bianchi et al., 2018). Relatedly, the findings are also consistent with theories of heuristic processing bias under conditions of adversity or cognitive depletion (Blessum et al., 1998; Sherman & Frost, 2000) and extend this work by showing that such processes may operate in the context of a macrolevel societal shock such as a pandemic. Results may also align with the greater prevalence and severity of the COVID-19 virus within Black communities and related media coverage, both possibly contributing to an uptick in discrimination (Chen et al., 2020; Woolf et al., 2021). Additional research will be needed to elucidate the specific reasons for the observed increases in discrimination and the relative contribution of each of these possible mechanisms. Irrespective of the underlying reasons for the reported findings, the results relating to discrimination are consistent with what has been referred to as double jeopardy for Black communities. That is, in addition to the greater physical toll of the virus, these communities have experienced a greater psychosocial toll during the pandemic vis-à-vis increases in discrimination and social isolation.

.128

.209



.214 (.086)\*

A581.13A\*\*

.461 (.083)\*\*

0.276 (.133)



Note. ISEL = Interpersonal Support Evaluation List; T = time. Path coefficients are unstandardized. Race is coded as White = 0 and Black = 1. All continuous variables are z scored, with T2 variables standardized by subtracting the T1 mean and dividing by the T1 SD to allow for parameter estimates to be interpreted with respect to changes over time in T1 SD units. Measurement error was specified by fixing the error variances of manifest variables for discrimination (EVD T1 and EVD T2) and social isolation (ISEL T1 and ISEL T2) using the formula (1-Reliability) × Sample Variance (Choi et al., 2011; Hayduk, 1987). Model fit:  $\chi^2$  (df = 4): .804, p = .938; standardized root mean square residual (SRMR) = .006. N = 263 See the online article for the color version of this figure.  $p^* < .05.$   $p^* < .01.$   $p^* < .001.$ 

More broadly, our results showing increases in social isolation for both Black and White students suggest that the pandemic has taken a toll on social relations. This finding builds on prior research showing greater increases in psychological distress during the pandemic in this age group as compared to middle aged and older adults (Pierce et al., 2020) and suggests that greater disruptions in social relations among young adults may be one potential explanation. The significance of the findings is also underscored by results of recent work showing that stress relating to social isolation during the pandemic is associated with increases in psychological distress among college students (Fruehwirth et al., 2021). The adverse effects of social isolation may also be greater in Black communities, where collectivism and spirituality are key cultural values (Hatter & Ottens, 1998; Neblett et al., 2010). The pandemic may have disrupted many key rituals related to these values (e.g., funerals, rites of passage, celebrations of key life events).

Social

Isolation T1

Race

.04

.07

ISEL T1

.162

Social isolation has been linked to mental health problems and a wide range of physical health outcomes including prospective associations with inflammation, heart disease, and mortality (Heffner et al., 2011; Smith et al., 2020; Yu et al., 2020). That discrimination was associated with increased disparities in social isolation-and partially mediated greater increases in social isolation among Black young adults-underscores the need for action to mitigate its pernicious consequences. Potential actions to address discrimination and its harmful effects have been discussed in detail elsewhere and include, for example, programs to increase belonging and support among the disproportionately stigmatized, interventions to reduce prejudice by increasing understanding and empathy, and broader efforts to reduce cultural racism and implicit bias (Paluck & Green, 2009; Williams & Cooper, 2019; Williams & Mohammed, 2013).

One unexpected finding was that those who reported higher levels of social isolation at baseline were more likely to experience an increase in discrimination. Although not hypothesized, this finding is not at odds with other hypotheses and results. Rather, it is consistent with notion that those who are part of more negatively stigmatized groups (of any variety, knowingly or knowingly) may be more likely to be socially isolated and also more likely to experience increases in discrimination during the pandemic. Although more data are needed to elucidate the directionality of associations between social isolation and discrimination, this finding suggests that a bidirectional or reciprocal relationship may be likely: Those with higher levels of social isolation may experience greater increases in discrimination in times of national stress, whereas concomitantly, increases in discrimination may in turn be associated with increases in social isolation. The possibility of mutually reinforcing linkages between social isolation and discrimination underscores the significance of our findings and calls for further research to understand the dynamics of a possibly cyclical relationship. Formulating and testing interventions to address such patterns are important directions for future inquiry.

.217 (.083)\*\*

Social

Isolation T2

Indirect effect = .100, SE = .046, p = .032Total effect = .376, SE = .132, p = .005

.563

ISEL T2

#### Limitations, Future Directions, and **Constraints on Generality**

Several limitations and future directions should be considered. Our interpretation of the findings was that observed changes in discrimination and social isolation likely occurred as a result of factors relating to the COVID-19 pandemic. However, a limitation of this study was that specific reasons for observed changes were not explicitly examined. Thus, alternative explanations for the findings are important to consider. One such alternative is that changes could 8

theoretically have been due to normative developmental changes in perceived discrimination or social isolation that occur in college. The first year of college in particular, for example, is thought to be a transitional time that could lead to such shifts. However, there is little evidence in previous research to suggest that large changes in perceived discrimination or social isolation normatively occur among college students (Fuller-Rowell et al., 2017). In one study that reported increases in perceived discrimination among minority students (Pinedo et al., 2021), effect sizes were small compared to those reported herein (approximately .1 SD unit increase across the first 2 years of college as compared to a .54 SD unit increase). Furthermore, supplemental analyses examining differences in our results between first-year students (51% of the sample) and more advanced students (49% of the sample) indicated no differences. These factors do not rule out the possibility of developmental change impacting the results, but do suggest that normative developmental change is unlikely to fully account for the reported findings. Additional research will be important to examine specific factors that may account for race differences in observed outcomes during the pandemic.

Another limitation of this study is that only one COVID-19 assessment was considered, relatively early in the pandemic. This early assessment is important for two key reasons: (a) the intensity of the initial lockdown and its psychosocial effects, and (b) because a small percentage of the population had been infected with COVID-19 at this stage, thus allowing for consideration of psychosocial effects with little contamination from direct physiological effects of the virus. However, the degree to which observed changes are enduring remains an open question. Furthermore, more robust mediation analyses can be conducted with additional waves of data (Cole & Maxwell, 2003; Roth & MacKinnon, 2011). A key direction of future research will therefore be to consider the time course of changes in measures of discrimination and social isolation—and related inequities—as the pandemic progressed.

The results of this study are not generalizable beyond the groups examined or methods used. Consideration of other demographic groups and measures will therefore be important next steps. For example, the degree to which similar findings would be evident in a national sample of Black and White adults or among other racial/ ethnic minority groups will be an important consideration to define the scope of the social problems implied by our findings. Consideration of variability across contexts (e.g., in more or less diverse residential environments), socioeconomic groups (e.g., among those with and without a college degree), and measures will also provide further insight. Our suspicion is that unfair treatment and discrimination directed toward stigmatized or disadvantaged groups may have increased broadly. Whether or not this is the case can be addressed empirically as more data become available.

Overall, the results of this study suggest notable shifts in experiences of discrimination and social isolation during the COVID-19 pandemic. Consistent with theories of heuristic processing bias in times of stress, the findings indicate that Black young adults experienced sizable increases in discrimination, which exacerbated racial disparities in social isolation. Determining the degree to which the reported changes are enduring and of longer term developmental significance within young adult populations—and whether similar changes are evident in other demographic groups—will be important next steps to inform a continuing pandemic response and responses to future societal-level stressful events.

#### References

- Acock, A. C. (2005). Working with missing values. Journal of Marriage and Family, 67(4), 1012–1028. https://doi.org/10.1111/j.1741-3737.2005.00191.x
- Anderson, D. M., Crost, B., & Rees, D. I. (2020). Do economic downturns fuel racial animus? *Journal of Economic Behavior & Organization*, 175, 9–18. https://doi.org/10.1016/j.jebo.2020.04.004
- Arnett, J. J., Žukauskienė, R., & Sugimura, K. (2014). The new life stage of emerging adulthood at ages 18–29 years: Implications for mental health. *The Lancet Psychiatry*, 1(7), 569–576. https://doi.org/10.1016/S2215-0366(14)00080-7
- Assari, S. (2018). Health disparities due to diminished return among Black Americans: Public policy solutions. *Social Issues and Policy Review*, *12*(1), 112–145. https://doi.org/10.1111/sipr.12042
- Beaney, T., Clarke, J. M., Jain, V., Golestaneh, A. K., Lyons, G., Salman, D., & Majeed, A. (2020). Excess mortality: The gold standard in measuring the impact of COVID-19 worldwide? *Journal of the Royal Society of Medicine*, *113*(9), 329–334. https://doi.org/10.1177/0141076 820956802
- Bianchi, E. C., Hall, E. V., & Lee, S. (2018). Reexamining the link between economic downturns and racial antipathy: Evidence that prejudice against blacks rises during recessions. *Psychological Science*, 29(10), 1584–1597. https://doi.org/10.1177/0956797618777214
- Blessum, K. A., Lord, C. G., & Sia, T. L. (1998). Cognitive load and positive mood reduce typicality effects in attitude–behavior consistency. *Personality and Social Psychology Bulletin*, 24(5), 496–504. https://doi.org/10 .1177/0146167298245005
- Bodenhausen, G. V. (1990). Stereotypes as judgmental heuristics: Evidence of circadian variations in discrimination. *Psychological Science*, 1(5), 319–322. https://doi.org/10.1111/j.1467-9280.1990.tb00226.x
- Bodenhausen, G. V. (1993). Emotions, arousal, and stereotypic judgments: A heuristic model of affect and stereotyping. In D. M. Mackie & D. L. Hamilton (Eds.), *Affect, cognition and stereotyping* (pp. 13–37). Academic Press. https://doi.org/10.1016/B978-0-08-088579-7.50006-5
- Chen, J. A., Zhang, E., & Liu, C. H. (2020). Potential impact of COVID-19related racial discrimination on the health of Asian Americans. *American Journal of Public Health*, 110(11), 1624–1627. https://doi.org/10.2105/ AJPH.2020.305858
- Choi, K.-H., Bowleg, L., & Neilands, T. B. (2011). The effects of sexism, psychological distress, and difficult sexual situations on U.S. women's sexual risk behaviors. *AIDS Education and Prevention*, 23(5), 397–411. https://doi.org/10.1521/aeap.2011.23.5.397
- Cohen, J., Cohen, P., Aiken, L. S., & West, S. G. (2003). Applied multiple regression/correlation analysis for the behavioral sciences (3rd ed.). Lawrence Erlbaum.
- Cohen, S., Mermelstein, R., Kamarck, T., & Hoberman, H. M. (1985). Measuring the functional components of social support. In I. G. Sarason & B. R. Sarason (Eds.), *Social support: Theory, research and applications* (pp. 73–94). Springer Netherlands., https://doi.org/10.1007/978-94-009-5115-0\_5
- Cole, D. A., & Maxwell, S. E. (2003). Testing mediational models with longitudinal data: Questions and tips in the use of structural equation modeling. *Journal of Abnormal Psychology*, 112(4), 558–577. https:// doi.org/10.1037/0021-843X.112.4.558
- Cole, D. A., & Preacher, K. J. (2014). Manifest variable path analysis: Potentially serious and misleading consequences due to uncorrected measurement error. *Psychological Methods*, 19(2), 300–315. https:// doi.org/10.1037/a0033805
- Dinenberg, R. E., McCaslin, S. E., Bates, M. N., & Cohen, B. E. (2014). Social support may protect against development of posttraumatic stress disorder: Findings from the Heart and Soul Study. *American Journal of Health Promotion*, 28(5), 294–297. https://doi.org/10.4278/ajhp.121023-QUAN-511
- Enders, C. K. (2010). Applied missing data analysis. Guilford Press.

- Enders, C. K. (2013). Dealing with missing data in developmental research. *Child Development Perspectives*, 7(1), 27–31. https://doi.org/10.1111/ cdep.12008
- Ernst, M., Niederer, D., Werner, A. M., Czaja, S. J., Mikton, C., Ong, A. D., Rosen, T., Brähler, E., & Beutel, M. E. (2022). Loneliness before and during the COVID-19 pandemic: A systematic review with meta-analysis. *American Psychologist*, 77(5), 660–677. https://doi.org/10.1037/amp0001005
- Ettman, C. K., Abdalla, S. M., Cohen, G. H., Sampson, L., Vivier, P. M., & Galea, S. (2020). Prevalence of depression symptoms in U.S. adults before and during the COVID-19 pandemic. *JAMA Network Open*, 3(9), Article e2019686. https://doi.org/10.1001/jamanetworkopen.2020.19686
- Fairlie, R. W. (2007). Explaining differences in access to home computers and the Internet: A comparison of Latino groups to other ethnic and racial groups. *Electronic Commerce Research*, 7(3), 265–291. https://doi.org/10 .1007/s10660-007-9006-5
- Faulkner, J., Schaller, M., Park, J. H., & Duncan, L. A. (2004). Evolved disease-avoidance mechanisms and contemporary xenophobic attitudes. *Group Processes & Intergroup Relations*, 7(4), 333–353. https://doi.org/ 10.1177/1368430204046142
- Fruehwirth, J. C., Biswas, S., & Perreira, K. M. (2021). The Covid-19 pandemic and mental health of first-year college students: Examining the effect of Covid-19 stressors using longitudinal data. *PLOS ONE*, *16*(3), Article e0247999. https://doi.org/10.1371/journal.pone.0247999
- Fuller-Rowell, T. E., Burrow, A. L., & Ong, A. D. (2011). Changes in racial identity among African American college students following the election of Barack Obama. *Developmental Psychology*, 47(6), 1608–1618. https:// doi.org/10.1037/a0025284
- Fuller-Rowell, T. E., Curtis, D. S., Doan, S. N., & Coe, C. L. (2015). Racial disparities in the health benefits of educational attainment: A study of inflammatory trajectories among African American and white adults. *Psychosomatic Medicine*, 77(1), 33–40. https://doi.org/10.1097/PSY .000000000000128
- Fuller-Rowell, T. E., Curtis, D. S., El-Sheikh, M., Duke, A. M., Ryff, C. D., & Zgierska, A. E. (2017). Racial discrimination mediates race differences in sleep problems: A longitudinal analysis. *Cultural Diversity and Ethnic Minority Psychology*, 23(2), 165–173. https://doi.org/10.1037/cdp0000104
- Fuller-Rowell, T. E., Evans, G. W., & Ong, A. D. (2012). Poverty and health: The mediating role of perceived discrimination. *Psychological Science*, 23(7), 734–739. https://doi.org/10.1177/0956797612439720
- Fuller-Rowell, T. E., Nichols, O. I., Robinson, A. T., Boylan, J. M., Chae, D. H., & El-Sheikh, M. (2021). Racial disparities in sleep health between Black and White young adults: The role of neighborhood safety in childhood. *Sleep Medicine*, 81, 341–349. https://doi.org/10.1016/j.sleep.2021.03.007
- Gable, S. L., & Bedrov, A. (2022). Social isolation and social support in good times and bad times. *Current Opinion in Psychology*, 44, 89–93. https:// doi.org/10.1016/j.copsyc.2021.08.027
- Garcia, M. A., Homan, P. A., García, C., & Brown, T. H. (2021). The color of COVID-19: Structural racism and the disproportionate impact of the pandemic on older Black and Latinx adults. *The Journals of Gerontology: Series B*, 76(3), e75–e80. https://doi.org/10.1093/geronb/gbaa114
- Gover, A. R., Harper, S. B., & Langton, L. (2020). Anti-Asian hate crime during the COVID-19 pandemic: Exploring the reproduction of inequality. *American Journal of Criminal Justice*, 45(4), 647–667. https://doi.org/10 .1007/s12103-020-09545-1
- Hamza, C. A., Ewing, L., Heath, N. L., & Goldstein, A. L. (2021). When social isolation is nothing new: A longitudinal study on psychological distress during COVID-19 among university students with and without preexisting mental health concerns. *Canadian Psychology*, 62(1), 20–30. https://doi.org/10.1037/cap0000255
- Hatter, D. Y., & Ottens, A. J. (1998). Afrocentric world view and Black students' adjustment to a predominantly White university: Does worldview matter? *College Student Journal*, 32(3), 472–480.
- Hayduk, L. A. (1987). Structural equation modeling with LISREL: Essentials and advances. JHU Press.

- Hayes, A. F., & Scharkow, M. (2013). The relative trustworthiness of inferential tests of the indirect effect in statistical mediation analysis: Does method really matter? *Psychological Science*, 24(10), 1918–1927. https://doi.org/10.1177/0956797613480187
- Heffner, K. L., Waring, M. E., Roberts, M. B., Eaton, C. B., & Gramling, R. (2011). Social isolation, C-reactive protein, and coronary heart disease mortality among community-dwelling adults. *Social Science & Medicine*, 72(9), 1482–1488. https://doi.org/10.1016/j.socscimed.2011.03.016
- Hope, E. C., Hoggard, L. S., & Thomas, A. (2015). Emerging into adulthood in the face of racial discrimination: Physiological, psychological, and sociopolitical consequences for african american youth. *Translational Issues in Psychological Science*, 1(4), 342–351. https://doi.org/10.1037/ tps0000041
- Hsiang, S., Allen, D., Annan-Phan, S., Bell, K., Bolliger, I., Chong, T., Druckenmiller, H., Huang, L. Y., Hultgren, A., Krasovich, E., Lau, P., Lee, J., Rolf, E., Tseng, J., & Wu, T. (2020). The effect of large-scale anticontagion policies on the COVID-19 pandemic. *Nature*, 584(7820), 262– 267. https://doi.org/10.1038/s41586-020-2404-8
- Kessler, R. C., Amminger, G. P., Aguilar-Gaxiola, S., Alonso, J., Lee, S., & Ustün, T. B. (2007). Age of onset of mental disorders: A review of recent literature. *Current Opinion in Psychiatry*, 20(4), 359–364. https://doi.org/ 10.1097/YCO.0b013e32816ebc8c
- Krieger, N., Smith, K., Naishadham, D., Hartman, C., & Barbeau, E. M. (2005). Experiences of discrimination: Validity and reliability of a selfreport measure for population health research on racism and health. *Social Science & Medicine*, 61(7), 1576–1596. https://doi.org/10.1016/j.socscimed .2005.03.006
- Krosch, A. R., Tyler, T. R., & Amodio, D. M. (2017). Race and recession: Effects of economic scarcity on racial discrimination. *Journal of Personality and Social Psychology*, *113*(6), 892–909. https://doi.org/10.1037/ pspi0000112
- Kusche, I., & Barker, J. L. (2019). Pathogens and immigrants: A critical appraisal of the behavioral immune system as an explanation of prejudice against ethnic outgroups. *Frontiers in Psychology*, 10, Article 2412. https://doi.org/10.3389/fpsyg.2019.02412
- Leigh-Hunt, N., Bagguley, D., Bash, K., Turner, V., Turnbull, S., Valtorta, N., & Caan, W. (2017). An overview of systematic reviews on the public health consequences of social isolation and loneliness. *Public Health*, 152, 157–171. https://doi.org/10.1016/j.puhe.2017.07.035
- Liu, Y., Finch, B. K., Brenneke, S. G., Thomas, K., & Le, P. D. (2020). Perceived discrimination and mental distress amid the COVID-19 pandemic: Evidence from the understanding America study. *American Journal of Preventive Medicine*, 59(4), 481–492. https://doi.org/10.1016/j .amepre.2020.06.007
- Lu, Y., Kaushal, N., Huang, X., & Gaddis, S. M. (2021). Priming COVID-19 salience increases prejudice and discriminatory intent against Asians and Hispanics. *Proceedings of the National Academy of Sciences of the United States of America*, 118(36), Article e2105125118. https://doi.org/10.1073/ pnas.2105125118
- Marmot, M., & Allen, J. (2020). COVID-19: Exposing and amplifying inequalities. *Journal of Epidemiology and Community Health*, 74(9), 681–682. https://doi.org/10.1136/jech-2020-214720
- Merz, E. L., Roesch, S. C., Malcarne, V. L., Penedo, F. J., Llabre, M. M., Weitzman, O. B., Navas-Nacher, E. L., Perreira, K. M., Gonzalez, F., Ponguta, L. A., Johnson, T. P., & Gallo, L. C. (2014). Validation of interpersonal support evaluation list-12 (ISEL-12) scores among Englishand Spanish-speaking Hispanics/Latinos from the HCHS/SOL Sociocultural Ancillary Study. *Psychological Assessment*, 26(2), 384–394. https:// doi.org/10.1037/a0035248
- Moran, J. B., Goh, J. X., Kerry, N., & Murray, D. R. (2021). Outbreaks and outgroups: Three tests of the relationship between disease avoidance motives and xenophobia during an emerging pandemic. *Evolutionary Psychological Science*, 7(4), 419–429. https://doi.org/10.1007/s40806-021-00283-z

- Muthen, L. K., & Muthen, B. O. (1998). Mplus user's guide (7th ed.).
- National Academies of Sciences, Engineering, and Medicine. (2021). *High and rising mortality rates among working-age adults*. The National Academies Press. https://doi.org/10.17226/25976
- Neblett, E. W., Jr., Hammond, W. P., Seaton, E. K., & Townsend, T. G. (2010). Underlying mechanisms in the relationship between Africentric worldview and depressive symptoms. *Journal of Counseling Psychology*, 57(1), 105–113. https://doi.org/10.1037/a0017710
- Newsom, J., Jones, R. N., & Hofer, S. M. (2013). Longitudinal data analysis: A practical guide for researchers in aging, health, and social sciences. Routledge. https://doi.org/10.4324/9780203814208
- Njoku, A. U. (2021). COVID-19 and environmental racism: Challenges and recommendations. *European Journal of Environment and Public Health*, 5(2), Article em0079. https://doi.org/10.21601/ejeph/10999
- O'Connor, R. C., Wetherall, K., Cleare, S., McClelland, H., Melson, A. J., Niedzwiedz, C. L., O'Carroll, R. E., O'Connor, D. B., Platt, S., Scowcroft, E., Watson, B., Zortea, T., Ferguson, E., & Robb, K. A. (2021). Mental health and well-being during the COVID-19 pandemic: Longitudinal analyses of adults in the UK COVID-19 Mental Health & Wellbeing study. *The British Journal of Psychiatry*, 218(6), 326–333. https://doi.org/ 10.1192/bjp.2020.212
- O'Shea, B. A., Watson, D. G., Brown, G. D. A., & Fincher, C. L. (2020). Infectious disease prevalence, not race exposure, predicts both implicit and explicit racial prejudice across the United States. *Social Psychological & Personality Science*, 11(3), 345–355. https://doi.org/ 10.1177/1948550619862319
- Paluck, E. L., & Green, D. P. (2009). Prejudice reduction: What works? A review and assessment of research and practice. *Annual Review of Psychology*, 60(1), 339–367. https://doi.org/10.1146/annurev.psych.60 .110707.163607
- Pan, S. W., Shen, G. C., Liu, C., & Hsi, J. H. (2021). Coronavirus stigmatization and psychological distress among Asians in the United States. *Ethnicity & Health*, 26(1), 110–125. https://doi.org/10.1080/13557858.2020.1849570
- Payne, T. J., Andrew, M., Butler, K. R., Wyatt, S. B., Dubbert, P. M., & Mosley, T. H. (2012). Psychometric evaluation of the interpersonal support evaluation list–short form in the ARIC study cohort. SAGE Open, 2(3), 1–8. https://doi.org/10.1177/2158244012461923
- Pazhoohi, F., & Kingstone, A. (2021). Associations of political orientation, xenophobia, right-wing authoritarianism, and concern of COVID-19: Cognitive responses to an actual pathogen threat. *Personality and Individual Differences*, 182, Article 111081. https://doi.org/10.1016/j.paid .2021.111081
- Peng, S., & Roth, A. R. (2022). Social isolation and loneliness before and during the COVID-19 pandemic: A longitudinal study of U.S. adults older than 50. *The Journals of Gerontology: Series B*, 77(7), e185–e190. https:// doi.org/10.1093/geronb/gbab068
- Perry, B. L., Aronson, B., & Pescosolido, B. A. (2021). Pandemic precarity: COVID-19 is exposing and exacerbating inequalities in the American heartland. *Proceedings of the National Academy of Sciences of the United States of America*, 118(8), Article e2020685118. https://doi.org/10.1073/ pnas.2020685118
- Pierce, M., Hope, H., Ford, T., Hatch, S., Hotopf, M., John, A., Kontopantelis, E., Webb, R., Wessely, S., McManus, S., & Abel, K. M. (2020). Mental health before and during the COVID-19 pandemic: A longitudinal probability sample survey of the U.K. population. *The Lancet Psychiatry*, 7(10), 883–892. https://doi.org/10.1016/S2215-0366(20)30308-4
- Pinedo, A., Durkee, M. I., Diemer, M. A., & Hope, E. C. (2021). Disentangling longitudinal trajectories of racial discrimination and critical action among Black and Latinx college students: What role do peers play? *Cultural Diversity and Ethnic Minority Psychology*, 27(3), 546–557. https://doi.org/10.1037/cdp0000434
- Priest, N., Paradies, Y., Trenerry, B., Truong, M., Karlsen, S., & Kelly, Y. (2013). A systematic review of studies examining the relationship between reported racism and health and wellbeing for children and young people.

Social Science & Medicine, 95, 115–127. https://doi.org/10.1016/j .socscimed.2012.11.031

- Rosenfeld, D. L., Balcetis, E., Bastian, B., Berkman, E. T., Bosson, J. K., Brannon, T. N., Burrow, A. L., Cameron, C. D., Chen, S., Cook, J. E., Crandall, C., Davidai, S., Dhont, K., Eastwick, P. W., Gaither, S. E., Gangestad, S. W., Gilovich, T., Gray, K., Haines, E. L., ... Torniyama, A. J. (2022). Psychological science in the wake of COVID-19: Social, methodological, and metascientific considerations. *Perspectives on Psychological Science*, *17*(2), 311–333. https://doi.org/10.1177/1745691621999374
- Roth, D. L., & MacKinnon, D. P. (2011). Mediation analysis with longitudinal data. In J. T. Newsom, R. N. Jones, & S. M. Hofer (Eds.), *Longitudinal data analysis: A practical guide for researchers in aging, health, and social sciences* (pp. 181–216). Routledge.
- Salari, N., Hosseinian-Far, A., Jalali, R., Vaisi-Raygani, A., Rasoulpoor, S., Mohammadi, M., Rasoulpoor, S., & Khaledi-Paveh, B. (2020). Prevalence of stress, anxiety, depression among the general population during the COVID-19 pandemic: A systematic review and meta-analysis. *Globalization and Health*, 16(1), Article 57. https://doi.org/10.1186/s12992-020-00589-w
- Schwartz, S. J. (2016). Turning point for a turning point: Advancing emerging adulthood theory and research. *Emerging Adulthood*, 4(5), 307–317. https://doi.org/10.1177/2167696815624640
- Serkez, Y. (2021, March 11). Opinion | we did not suffer equally. New York Times. https://www.nytimes.com/interactive/2021/03/11/opinion/covidinequality-race-gender.html
- Sherman, J. W., & Frost, L. A. (2000). On the encoding of stereotype-relevant information under cognitive load. *Personality and Social Psychology Bulletin*, 26(1), 26–34. https://doi.org/10.1177/0146167200261003
- Smith, K. J., Gavey, S., RIddell, N. E., Kontari, P., & Victor, C. (2020). The association between loneliness, social isolation and inflammation: A systematic review and meta-analysis. *Neuroscience and Biobehavioral Reviews*, 112, 519–541. https://doi.org/10.1016/j.neubiorev.2020.02.002
- Solmi, M., Radua, J., Olivola, M., Croce, E., Soardo, L., Salazar de Pablo, G., Il Shin, J., Kirkbride, J. B., Jones, P., Kim, J. H., Kim, J. Y., Carvalho, A. F., Seeman, M. V., Correll, C. U., & Fusar-Poli, P. (2022). Age at onset of mental disorders worldwide: Large-scale meta-analysis of 192 epidemiological studies. *Molecular Psychiatry*, 27(1), 281–295. https://doi.org/ 10.1038/s41380-021-01161-7
- Steptoe, A., Shankar, A., Demakakos, P., & Wardle, J. (2013). Social isolation, loneliness, and all-cause mortality in older men and women. *Proceedings of the National Academy of Sciences of the United States of America*, 110(15), 5797–5801. https://doi.org/10.1073/pnas.1219686110
- Stok, F. M., Bal, M., Yerkes, M. A., & de Wit, J. B. F. (2021). Social inequality and solidarity in times of COVID-19. *International Journal of Environmental Research and Public Health*, 18(12), Article 6339. https:// doi.org/10.3390/ijerph18126339
- Su, Z., McDonnell, D., Ahmad, J., Cheshmehzangi, A., Li, X., Meyer, K., Cai, Y., Yang, L., & Xiang, Y.-T. (2020). Time to stop the use of 'Wuhan virus,' 'China virus' or 'Chinese virus' across the scientific community. *BMJ Global Health*, 5(9), Article e003746. https://doi.org/10.1136/bmjgh-2020-003746
- Tajfel, H., & Turner, J. (2001). An integrative theory of intergroup conflict. In M. A. Hogg & D. Abrams (Eds.), *Intergroup relations: Essential readings* (pp. 94–109). Psychology Press.
- Taylor, T. R., Kamarck, T. W., & Shiffman, S. (2004). Validation of the Detroit Area Study Discrimination Scale in a community sample of older African American adults: The Pittsburgh healthy heart project. *International Journal of Behavioral Medicine*, 11(2), 88–94. https://doi.org/10 .1207/s15327558ijbm1102\_4
- Williams, D. R., & Cooper, L. A. (2019). Reducing racial inequities in health: Using what we already know to take action. *International Journal* of Environmental Research and Public Health, 16(4), Article 606. https:// doi.org/10.3390/ijerph16040606
- Williams, D. R., & Mohammed, S. A. (2009). Discrimination and racial disparities in health: Evidence and needed research. *Journal of Behavioral Medicine*, 32(1), 20–47. https://doi.org/10.1007/s10865-008-9185-0

- Williams, D. R., & Mohammed, S. A. (2013). Racism and health II: A needed research agenda for effective interventions. *American Behavioral Scientist*, 57(8), 1200–1226. https://doi.org/10.1177/0002764 213487341
- Williams, D. R., Yu, Y., Jackson, J. S., & Anderson, N. B. (1997). Racial differences in physical and mental health: Socio-economic status, stress and discrimination. *Journal of Health Psychology*, 2(3), 335–351. https:// doi.org/10.1177/135910539700200305
- Woolf, S. H., Masters, R. K., & Aron, L. Y. (2021). Effect of the covid-19 pandemic in 2020 on life expectancy across populations in the USA and other high income countries: Simulations of provisional mortality data. *BMJ*, 373(1343), Article n1343. https://doi.org/10.1136/bmj.n1343
- Wu, C., Qian, Y., & Wilkes, R. (2021). Anti-Asian discrimination and the Asian-white mental health gap during COVID-19. *Ethnic and Racial Studies*, 44(5), 819–835. https://doi.org/10.1080/01419870.2020.1851739
- Yu, B., Steptoe, A., Chen, L.-J., Chen, Y.-H., Lin, C.-H., & Ku, P.-W. (2020). Social isolation, loneliness, and all-cause mortality in patients with cardiovascular disease: A 10-year follow-up study. *Psychosomatic Medicine*, 82(2), 208–214. https://doi.org/10.1097/PSY.000000000000777

Received November 24, 2021

Revision received July 20, 2022

Accepted July 22, 2022 ■